Author’s response to reviews

Title: Systemic and tumor-directed therapy for oligometastatic prostate cancer: study protocol for a phase II trial for Veterans with de novo oligometastatic disease

Authors:

Neil Parikh (nrparikh@mednet.ucla.edu)
Claudia Huiza (claudia.huiza@va.gov)
Jill Patel (jillpatel@mednet.ucla.edu)
Sonny Tsai (Sonny.Tsai@va.gov)
Nathisha Kalpage (nkalpage@mednet.ucla.edu)
May Thein (may.thein@va.gov)
Sage Pitcher (sage.pitcher@va.gov)
Steve Lee (stevepl@uci.edu)
Warren Inouye (warren.inouye@va.gov)
Mark Jordan (Mark.Jordan2@va.gov)
Homayoon Sanati (Homayoon.Sanati@va.gov)
Lida Jafari (lida.jafari@va.gov)
Carol Bennett (carol.bennett@va.gov)
Greg Gin (Greg.Gin@va.gov)
Amar Kishan (aukishan@mednet.ucla.edu)
Robert Reiter (rreiter@mednet.ucla.edu)
Michael Lewis (michael.lewis8@va.gov)
Ahmad Sadeghi (ahmad.sadeghi@va.gov)
William Aronson (waronson@g.ucla.edu)
Isla Garraway (igarraway@mednet.ucla.edu)
Matthew Rettig (matthew.rettig@va.gov)

Nicholas Nickols (nnickols@mednet.ucla.edu)

Version: 2 Date: 12 Mar 2019

Author’s response to reviews:

1. Please remove the SPIRIT checklist and the second copy of the manuscript from the manuscript file.

We removed the SPIRIT checklist and the tracked changes copy of the manuscript.

2. Please, include the Trial Registration section in the abstract provided in the Editorial Manager System.

The abstract now has the clinicaltrials.gov registration in the Trial Registration section.

3. Please, include a section "Trial design" in the Methods section.

There is a section “Trial design” in the Methods section now.

4. Thank you for providing the SPIRIT checklist and for adhering to the guidelines.

You are welcome.

5. Please, split the section "Funding acknowledgement" in 2 separate sections: Funding and Acknowledgement. In the Funding declaration only information about the funding for the study should be displayed. The information "NN is also a Prostate Cancer Foundation Young Investigator" should be included under Author's Information section.

We split the section "Funding acknowledgement" in 2 separate sections: Funding and Acknowledgement. The information "NN is also a Prostate Cancer Foundation Young Investigator" is included under Author's Information section.

6. In the Authors' contribution section, please clarify what is the meaning of this contribution: "Participating centers: AS, MJ, GG, ML, SL, WI, RJ, HS". The individual contributions of all authors to the manuscript should be specified in the Authors’ Contributions section.

We have clarified the authors’ contribution section.

7. As it is stated in the manuscript that "Patients are recruited from the urology, medical oncology, and radiation oncology clinics at the VA Greater Los Angeles and VA Long Beach Healthcare Systems.", could you please clarify whether the ethical approval you provided covers all the study sites and send evidence of these approvals to the following email address:
BMCSeriesEditorial@biomedcentral.com. We kindly ask you to provide the original and the translated copy for all documents.

We emailed the VA Long Beach approval documents to BMCSeriesEditorial@biomedcentral.com. We already emailed the VA Greater Los Angeles approval documents.

Best,
Nick